Author's response to reviews

Title: Multicenter phase II study of weekly docetaxel, cisplatin, and S-1 (TPS) induction chemotherapy in locally advanced squamous cell cancer of the head and neck

Authors:

Woo Kyun Bae (drwookyun@jnu.ac.kr)
Jun Eul Hwang (hiunyl@naver.com)
Hyun Jeong Shim (hjnhj@hanmail.net)
Sang Hee Cho (shcho@jnu.ac.kr)
Ki Hyeong Lee (kihlee@chungbuk.ac.kr)
Hye Suk Han (sook3529@hanmail.net)
Euk-Kee Song (eksong@jbnu.ac.kr)
Hwan Jung Yun (hjyun@cnu.ac.kr)
In Sung Cho (cis3379@dreamwiz.com)
Joon Kyoo Lee (joonkyoo@jnu.ac.kr)
Sang-Chul Lim (limsc@jnu.ac.kr)
Woong-Ki Chung (wkchung@jnu.ac.kr)
Ilk-Joo Chung (ijchung@jnu.ac.kr)

Version: 2 Date: 20 July 2012

Author's response to reviews: see over
Dear Editor,

Please accept for consideration of publication in *BMC Cancer* the manuscript titled “Multicenter phase II study of weekly docetaxel, cisplatin, and S-1 (TPS) induction chemotherapy in locally advanced squamous cell cancer of the head and neck (NCT01645748)” by Woo Kyun Bae, Jun Eul Hwang, Hyun Jeong Shim, Sang Hee Cho, Ki Hyeong Lee, Hye Suk Han, Eun-Kee Song, Hwan Jung Yun, In Sung Cho, Joon Kyoo Lee, Sang-Chul Lim, Woong-Ki Chung and Ik-Joo Chung.

All authors have read and approved submission of the manuscript and the manuscript has not been published and is not being considered for publication elsewhere in whole or part in any language.

I continue to thank yourself, your staff for the time and effort you all spend on my behalf.

Please address all correspondence to myself.

Sincerely yours,

Sang Hee Cho MD, PhD
Department of Internal Medicine
Chonnam National University Medical School
Hakdong, Dongku, 501-746
Gwangju, South Korea
Phone: +82-61-379-7633
Fax: +82-61-379-8019
E-mail: shcho@jnu.ac.kr